Author's response to reviews

Title: Algorithms for the diagnosis and treatment of restless legs syndrome in primary care

Authors:

Diego Garcia-Borreguero (dgb.eurlssg@gmail.com)
Paul Stillman (drefdry@yahoo.fr)
Heike Benes (drefdry@gmail.com)
Heiner Buschmann (drefdry@gmail.com)
Ray Chaudhuri (drefdry@gmail.com)
Victor M Gonzalez (drefdry@gmail.com)
Birgit Högl (drefdry@gmail.com)
Ralf Kohnen (drefdry@gmail.com)
Giorgio Carlo Monti (drefdry@gmail.com)
Karin Stiasny-Kolster (drefdry@gmail.com)
Claudia Trenkwalder (drefdry@gmail.com)
Anne-Marie Williams (msrx4amw@gmail.com)
Marco Zucconi (drefdry@gmail.com)

Version: 2 Date: 29 November 2010

Author's response to reviews:

As requested we have merged the two manuscripts into one. We do now hope that it is acceptable for publication in BMC Neurology.

Kind regards,

D Garcia-Borreguero